Author’s response to reviews

Title: Universal goniometer and electro-goniometer intra-examiner reliability in measuring the knee range of motion during active knee extension test in patients with chronic low back pain with short hamstring muscle

Authors:

MohammadBagher Shamsi (mbshamsi@yahoo.com)

Maryam Mirzaei (mirzaei.m.epid92@gmail.com)

Seyyed Saeed Khabiri (saeed.khabiri@gmail.com)

Version: 4 Date: 03 Mar 2019

Author’s response to reviews:

1. We note that you have included a ‘Consent for publication’ section in the Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity.

Please state in this section if you obtained written consent from the patient in figure 2 to publish this image.

-Respond to editor comment 1: Amended in the manuscript.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

-Respond to editor comment 2: Amended in the manuscript.

Technical Comments:
(1) Please ensure that you have cited all your figures and tables in the main text; currently table 1 is not cited.

- Respond to editor comment: Amended in the manuscript.